



Article Circular Bioeconomy in the Metropolitan Area of Barcelona: Policy Recommendations to Optimize Biowaste Management

Karin Meisterl^{1,*}, Sergio Sastre¹, Ignasi Puig-Ventosa¹, Rosaria Chifari¹, Laura Martínez Sánchez², Laurène Chochois³, Gabriella Fiorentino⁴ and Amalia Zucaro⁴

- ¹ Fundació ENT, c/Josep Llanza, 1-7, 2n 3a, 08800 Vilanova i la Geltrú, Spain; ssastre@ent.cat (S.S.); ipuig@ent.cat (I.P.-V.); rchifari@ent.cat (R.C.)
- ² BR, Barcelona Regional, Agència de Desenvolupament Urbà S.A., Carrer 60, 25-27. Edifici Z. 2a Planta, Sector A, Zona Franca, 08040 Barcelona, Spain; laura.martinez@bcnregional.com
- ³ LIST, Luxembourg Institute of Science and Technology, 5 Av. des Hauts-Fourneaux, 4362 Esch-Belval Esch-sur-Alzette, Luxembourg; laurene.chochois@list.lu
- ⁴ ENEA, Division Resource Efficiency, Department for Sustainability, Research Centre of Portici, 80055 Portici, Italy; gabriella.fiorentino@enea.it (G.F.); amalia.zucaro@enea.it (A.Z.)
- * Correspondence: kmeisterl@ent.cat

Abstract: Municipal biowaste management is at the core of the transition towards a circular bioeconomy in the EU. However, most urban systems are still far from being aligned with these principles. This paper addresses the case of the Metropolitan Area of Barcelona. The current system of biowaste management is compared with a more sustainable alternative scenario. Regulatory and non-regulatory drivers and barriers for the transition from the current state to the alternative scenario are identified and later transformed into policy recommendations using a multi-stakeholder approach. This paper focuses on the separate collection of biowaste and the production of biomethane. Increasing the quantity and quality of separate biowaste collection is a prerequisite for the market-relevant production of biogas from anaerobic digestion that can be converted into biomethane. The results show that more efficient collection systems such as door-to-door or smart bins together with tax incentives such as the pay-as-you-throw principle are key to increasing the amount of collected biowaste, while targeted communication combined with controls and penalties are key to minimizing impurities. In addition to financial incentives for the construction of new anaerobic digestion plants, financial incentive systems are also required for the biomethane sector to ensure competitiveness with fossil fuels.

Keywords: circular bioeconomy; separate biowaste collection; biogas; biomethane; biowaste valorization; drivers and barriers; policy recommendations

1. Introduction

About 113.8 Mt of municipal biowaste per year is generated in Europe (EU27+), of which about 60 Mt is food waste [1]. Over the past two decades, European Union (EU) policy-makers have given high priority to incentives and directives aimed at reducing the amount of biowaste and improving the separate biowaste collection, treatment, and valorization. The most significant ones are the European Union (EU) Waste Framework Directive [2], the European Commission (EC) Circular Economy Action Plan [3], and the European Commission (EC) Bioeconomy strategy [4] and its Action Plan [5]—a core part of the European Green Deal [6]. Even if the 'umbrella' concepts green economy, circular economy, and bioeconomy are based on different operational strategies, they are united by the common ideal of linking economic, ecological, and social issues. In particular, the resource-focused bioeconomy and circular economy concepts are strongly connected, but there is much potential to make bio-based and bio-ecological strategies more circular [7]. While a circular economy aims at increasing the resource efficiency of processes and



Citation: Meisterl, K.; Sastre, S.; Puig-Ventosa, I.; Chifari, R.; Martínez Sánchez, L.; Chochois, L.; Fiorentino, G.; Zucaro, A. Circular Bioeconomy in the Metropolitan Area of Barcelona: Policy Recommendations to Optimize Biowaste Management. *Sustainability* **2024**, *16*, 1208. https://doi.org/ 10.3390/su16031208

Academic Editors: Alberto Bezama and Nora Szarka

Received: 13 November 2023 Revised: 12 January 2024 Accepted: 22 January 2024 Published: 31 January 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the use of recycled materials to reduce material consumption [8], a bioeconomy aims at promoting sustainable production of natural resources from biomass rather than fossiland mineral-based resources. The biogenic resources (i.e., agriculture, forestry, fisheries, and aquaculture) can, in turn, become feedstock for the economic and industrial sectors to produce: (1) food and feed, (2) fibers, (3) chemical and materials that later will be transformed into bio-based products, and (4) bioenergy and biofuels substituting fossil feedstock [4]. As far as the risk of a linear bioeconomy is concerned (competition of land uses, damage to soils, energy demand, fertilizer consumption, etc.), the bioeconomy offers many promising alternatives for replacing fossil fuel products and services, but it also has many limitations. Circular economy and bioeconomy are, therefore, complementary concepts that can reinforce each other. A circular bioeconomy (CBE) unites the main strengths of both concepts and focuses on reducing the use of petrochemicals, mitigating climate change, reducing dependence on natural resource imports, and boosting local economies [9].

Secondary biomass, i.e., biowaste, organic by-products and residues, plays a key role in CBE strategies. As a priority of the EU waste hierarchy [2], waste generation—including food waste as a major part of municipal biowaste—should be reduced. In the case of unavoidable organic waste and residues, it can be considered as a resource that can be reused, recovered, or recycled. The EU Biorefinery Forecast to 2030 [10] supports the EU Bioeconomy Action Plan [5] to promote the development and use of new resourceefficient and sustainable biorefineries in Europe. Biorefining can be broadly defined as the processing of biomass into a range of marketable bio-based products, such as food and feed, chemicals and materials, and bioenergy, through thermochemical, physicochemical, and biochemical processes [11]. With bioproducts and bio-based value chains as the central pathways to a resource-efficient circular economy, this "valorization and value chains" approach encompasses a complex web of processes and actors that lead to socio-economic growth, environmental benefits, and technological progress [12].

This paper shows an excerpt of the results of the BBI-JU H2020 project BIOCIR-CULARCITIES (BCC). BCC concentrates on the CBE potential in cities and, therefore, rethinking selected infrastructure systems and supply chains, unlocking the potential of unexploited bio-based waste. BCC investigates the development of economically and environmentally sustainable organic waste valorization models in three pilot territories, each covering different bio-based value chains: (i) forestry residues in the Province of Pazardzhik, Bulgaria; (ii) agro-industrial biowaste (coffee silverskin) in the Metropolitan City of Naples, Italy; and (iii) municipal biowaste in the Metropolitan Area of Barcelona, Spain.

For each of the pilot territories, BCC explored current organic waste management models hand in hand with local stakeholders in order to define an alternative scenario in line with the CBE principles [13]. Both the current situation and the alternative scenarios were compared using life cycle assessment (LCA) and life cycle costing (LCC) to ensure that the proposed alternative scenarios are environmentally friendly, economically feasible, and improve the current management systems [14,15]. In parallel, a multi-actor approach (i.e., including stakeholders and external experts) was carried out for identifying legal, technical, economic, environmental, and social drivers and barriers to a shift from the current to the alternative scenarios [16]. Finally, policy recommendations to promote that shift were formulated [17]. In this publication, we focus on the holistic process needed to identify drivers and barriers and provide policy recommendations to implement a more sustainable municipal biowaste management in the Metropolitan Area of Barcelona (MAB) case study.

The MAB case study aims at improving the quantity and quality of separately collected biowaste to be treated in an anaerobic digestion (AD) plant and subsequently increasing biogas production to be upgraded into biomethane. Biomethane was chosen as the end product of the value chain studied because it offers the possibility to diversify energy sources and reduce dependence on imported natural gas [18]. It also has great potential to help decarbonize the energy sector and can, therefore, play a crucial role in reducing greenhouse gas (GHG) emissions and contribute to the Paris Agreement [19]. When derived from waste sources, using biomethane can result in a net reduction in carbon dioxide (CO₂) emissions [20].

This paper addresses two main research questions: (1) What are the current drivers and barriers—based on the current legal framework and stakeholder feedback—for the successful implementation of a circular economy focusing on separate collection improvement and biogas/biomethane production in the MAB., and (2) what policy recommendations can support the implementation of the proposed alternative value chain in the MAB in line with the bioeconomy principles.

The article is structured into six sections. Section 1 contextualizes the research and provides background. Section 2 presents the materials and methods of the research. Section 3 shows the results in terms of main drivers and barriers for the implementation of a scenario in which municipal biowaste management is improved all along the value chain, focusing on separate collection and biomethane production. Section 4 discusses drivers, barriers, and its transformation into policy recommendations. Section 5 discusses the results. Section 5 concludes the findings and provides directions for future research.

2. Materials and Methods

The detection of regulatory and non-regulatory drivers and barriers to shift from the current to the alternative scenario and the provision of policy recommendations to achieve this transition have been structured in a comprehensive process combining research desk and multi-actor (MA) involvement. The entire process is structured in two main groups of tasks, both including MA engagement (Figure 1). The first set of tasks is related to the technical definition and quantification of the current municipal biowaste management scenario at the different stages (e.g., biowaste generation, collection, treatment, etc.). In addition, this first part of the comprehensive process includes activities related to the design of an alternative management scenario aligned with CBE principles that is environmentally sound according to LCA [21] and economically feasible according to LCC [22].



Figure 1. Schematic representation of the methodological steps of the comprehensive process to identify drivers, barriers, and policy recommendations combing research desk and multi-actor (MA) involvement. MA indicates the specific steps where the multi-actor approach was applied, involving local stakeholders and external experts. Source: Own elaboration.

The second group is related to the proper identification of drivers and barriers and formulation of policy recommendations. Drivers and barriers were, in turn, mapped in a

two-step exercise: (1) gathering and analyzing information from a research desk analysis and then (2) collecting and elaborating data from stakeholder participation. First, the drivers and barriers, arising from the legal framework, were identified on the basis of an expert analysis of the most important policy framework documents at EU, national, regional, and local level. During this step, international good practices (GPs), related to the selected MAB pilot area, were compiled to later support policy recommendations. Secondly, the drivers and barriers were identified in dedicated meetings with local stakeholders and external international experts not associated with the BCC consortium. Departing from the drivers and barriers identified, a set of policy recommendations were developed and, where possible, illustrated with best practices found in the previous step. Each of the methodological steps are further elaborated in the following sections.

2.1. Multi-Actor Involvement

Building a shared vision of the current municipal biowaste management system, validating alternative scenarios, co-creating common understanding of drivers and barriers, and formulating relevant policy recommendations, requires stakeholders' involvement, for which an MA approach according to the Quadruple Helix Model was set up. The Quadruple Helix Model is an efficient way of fostering cooperation in research and development between four major actors in the innovation system: science, policy, industry, and society [23]. The real added value of involving multiple actors in the decision-making process is to develop "sound and fair" policies by incorporating "useful" (relevant and sound) information into the governance process and by increasing the transparency and quality of the evaluation and decision-making process, taking into account the individual interests of the different social actors involved [24].

Stakeholder involvement took place, on the one hand, through four bi-annual Local Living Labs on-site and online in the language of the MAB (Catalan), attended by relevant local stakeholders related to the municipal biowaste management value chain. In addition, each local Living Lab was followed by an online advisory board meeting and an on-site or online peer review session (both in English) with international experts.

The external project stakeholders involved in the MAB pilot site were:

- 1. Local stakeholders: 19 stakeholders including local waste management authorities, other local authorities, academic and research institutions, and private companies involved in municipal biowaste management. These stakeholders helped to (i) validate the information related to the state of the art of the current biowaste management scenario, (ii) select the most realistic and sustainable alternative scenario, (iii) identify the related existing drivers and challenges in the current biowaste management system and for the implementation of the alternative scenario, and (iv) give feedback on policy recommendations for a successful implementation of the alternative scenario.
- 2. International external experts: 18 external experts were selected to provide input after each round of Living Labs. For these tasks, experts from academic and research institutions, research clusters, environmental consultancy companies and engineering companies were selected. Their contribution was key to validate and support internal decision-making and preliminary results, as well as to answer the specific technical questions prepared by the consortium.

2.2. Research Desk Analysis: Elaboration of the Policy Framework Database, Identification of GPs, and Identification of Regulatory Drivers and Barriers to the MAB Alternative Scenario

A database including the most relevant documents regarding the EU policy framework was compiled addressing the EU level plus the national, regional, and local level in the MAB. Departing from the initial database including 95 documents, a first relevance analysis was carried out from which a shorter list of references (47 documents) was filtered for an in depth analysis to extract regulatory drivers and barriers to the implementation of CBE alternative scenario, including 23 documents at the EU level, 12 at the national level (Spain), 10 at the regional level (Catalonia), and 2 at the local level (MAB).

Later, these documents were systematically analyzed to map relevant passages including explicit legal drivers and barriers that might foster or hinder the implementation of alternative scenarios to the current biowaste management value chains. In this paper, "drivers" are defined as all factors that favor the implementation of the CBE and, therefore, the alternative scenario in a bio-based value chain, whereas "barriers" are factors that hinder its implementation. Drivers and barriers were classified into 6 categories according to the scope they were referred to:

- **Legal/Administrative (L/A):** EU-wide, national, regional, and local legislation and its implementation, e.g., through binding targets, administrative penalties for non-compliance, etc.
- **Technical (Tec):** Technological innovations, best available techniques (BAT), available infrastructure and equipment, waste quality requirements, etc.
- Economic (Ec): Capacity for investment in innovation, BAT and new product chains, profitability threshold, market situation of raw materials and bio-based products, economic incentives such as tax reductions, waste charges/fees and subsidies.
- Environment/Health (E/H): Greenhouse gas emissions, possible other impacts on environment and health.
- Social (Soc): Public acceptance of new biowaste collection systems and bio-based products, awareness raising campaigns on biowaste separate collection or biowastebased products.
- **Stakeholder involvement (SI):** Involvement of various stakeholders from science, industry, politics, citizens, and NGOs in decision-making processes; interdisciplinary exchange via stakeholder platforms.

The identified GPs related to CBE belong to different databases [25]. The very first criterion for selecting the GPs was to focus on actions related to the specific pilot areas. First, existing GPs in Spain and Catalonia to be transferred to the MAB pilot were screened in the "Ministry for the Ecological Transition and Demographic Challenge" (MITERD) platform, the Catalan "Residus Municipals" (municipal waste) platform, and in the "Som gent de profit" (a Catalan platform devoted to prevention of food wastage). In addition, at the European level, the "European Circular Economy Stakeholder Platform" and the "Municipal Waste Europe" platform were considered. During the first local Living Lab, together with the local partners, the main GPs were selected for the three project pilot areas and new ones were proposed. Further proposals came from international experts during the third peer review session. For the MAB pilot, the most important criterion for the selection of GPs was their focus on the optimization of the quality and quantity of biowaste to be used in biocircular biorefinery processes.

2.3. Status Quo Analysis of the Current Municipal Biowaste Management in the Metropolitan Area of Barcelona

The MAB is a territorial entity located in Catalonia, in the province of Barcelona in northeastern Spain (Figure 2). The MAB comprised 36 municipalities with 3.3 million inhabitants in 2021 [26], including the city of Barcelona, which means that 43% of Catalonia's population lives in less than 2% of the region's area, ranking MAB as the eighth most populated metropolitan area in the EU. This implies a very high population density, settled predominantly in urban areas interspersed with agricultural and Mediterranean forests and mountain areas. The MAB accounts for 52% of Catalonia's GDP.

The definition and quantification of the current municipal biowaste management scenario was developed with data provided by the local authority of the MAB pilot area. The MAB pilot area has a total of four waste treatment plants called "Ecoparcs" and two industrial composting plants where biowaste separately collected in the MAB is sent. Ecoparc 2 (Montcada i Reixac) was selected as pilot study because it treats significant amounts of municipal biowaste separately collected in the MAB, and also because the upgrading of biogas to biomethane could be implemented in the short-term.



Figure 2. Location of the BCC pilot study area: the Metropolitan Area of Barcelona in the region of Catalonia in north-eastern Spain (highlighted in red). Source: Own elaboration.

Municipal waste management in the MAB is carried out so that five main waste streams are collected from both households and commercial facilities. The main collection system comprises the separate collection of glass packaging, lightweight packaging, paper and cardboard, food biowaste (kitchen organic waste), and garden waste. Along with residual waste, in general, these streams are collected through open street bins. In addition, bulky waste is collected separately by request, and there is a network of "deixalleries" (recycling centers) where other waste streams are collected separately (e.g., WEEE, mineral oils, etc.). Furthermore, specific separate collection services are provided to commercial activities (e.g., for paper and cardboard).

According to the Spanish regulatory framework, municipal waste collection and treatment are mandatory services to be provided by local authorities [27]. These services can be provided either individually (i.e., each municipality establishes its own collection and treatment services and facilities) or in coordination with other municipalities. In general, in Spain and in Catalonia, the most frequent municipal waste management configuration entails the provision of waste collection services by each municipality while treatment is provided by a local supra-municipal entity, so treatment infrastructures are shared to create economies of scale and minimize treatment and disposal costs.

In 2021 (year of current scenario), separately collected municipal biowaste and green waste represented 13.5% (199,628 t) of the total municipal solid waste (MSW) generation in the MAB (1,478,128 t). The total amount of generated biowaste was estimated to be 431,310 t, meaning that currently, only 46% of the total potential are separately collected. There are three main sources of separately collected biowaste in the MAB. First, kitchen and garden waste (small size green waste) from households and small businesses which is normally disposed of in open 2200-litre street containers and collected by municipal waste collected separately. The third municipal biowaste stream is green waste from municipal park operations. In total, 74% of the total separately collected biowaste was household biowaste and it includes 15% of impurities.

In 2021, Ecoparc 2 received around 43% (85,640 t) of the separately collected biowaste in the MAB: 77,244 t from municipal biowaste plus 8396 t from commercial or industrial sources for treatment. No green waste from municipal park operations is sent to Ecoparc 2. Figure 3 shows the current management system for separately collected biowaste sent to Ecoparc 2.

The treatment steps include the pretreatment biowaste to remove non-organic materials (impurities). After pretreatment, the biowaste entered AD. The products of AD include biogas and digestate (solid + liquid). Digestate is treated as follows: the liquid digestate is

sent to wastewater treatment where some of the solid digestate is incinerated and some is sent to other specific treatments/uses. The biogas produced is converted into electricity by means of a combined heat and power (CHP), of which 60% is fed into the national electricity grid, whereas the rest (40%) is consumed by the AD plant. A part of the solid digestate obtained from AD becomes compost.



Figure 3. Scheme of the current municipal biowaste management system in the Ecoparc 2 (Current scenario). Source: Own elaboration based on data provided by the MAB local authority and data from DECISIVE database [28].

2.4. Assessment of the Alternative Scenario for Municipal Biowaste in the MAB

The alternative scenario was designed along with representatives from the local authorities Àrea Metropolitana de Barcelona (AMB) and Barcelona Regional (BR), plus several other stakeholders involved in the management of the municipal biowaste treatment plant "Ecoparc 2" [13] and proved to be more sustainable and more economically advantageous than the current situation [14,15]. Local actors suggested to consider a more feasible, realistic, and short-term (timeframe of five years from 2021) alternative scenario instead of hypothesizing innovative technologies. This was mainly due to the scaling up limitations of prototype processes that, while they may have been more innovative, would have been difficult to actualize.

The alternative scenario entails improvements in the three main stages of the value chain including biowaste prevention, separate collection of biowaste, and biowaste treatment: (i) the introduction of prevention measures to reduce 5% of the current amount of biowaste; (ii) a change in the separate collection of biowaste from open street containers to door-to-door (DtD) collection and/or smart bins to increase the quality and quantity of separately collected biowaste; and (iii) the upgrading of the biogas produced from AD into biomethane to be injected into the local gas grid or to be used as biofuel. Mass and energy flows in the alternative scenario are shown in Figure 4. The biowaste reduction (5% of total input biowaste to Ecoparc 2) due to prevention measures and the increase in separately collected biowaste (30% of total input biowaste) were first discussed and then validated with local stakeholders. The combination of these targets will result in a higher quality of source-separated biowaste (less contamination), which in turn will result in higher quality compost, higher biogas production, and a reduction in waste destined for landfill or incineration within the short time horizon of five years.



Figure 4. Scheme of the alternative biowaste management scenario in the Ecoparc 2 (Alternative Scenario). Source: Own elaboration based on data provided by the MAB local authority and data from DECISIVE database [28].

In contrast to the current situation, higher volumes of biogas would have been obtained and purified into biomethane by a mix of the most common technologies, namely amino washing, membrane technique, pressure swing adsorption in addition to biomethane production from synthetic gas by fluidized or fixed bed technology. A production of $8,012,157 \text{ m}^3$ of biomethane was estimated. For the alternative scenario, it was calculated that the entire biogas production is upgraded into biomethane. Electricity and heat requirements of the AD plant are assumed to be drawn from the national grid. For the calculation of the biomethane upgrading, a methane (CH₄) content of 56% in the biogas after AD is assumed. In total, 40,113 t of the digestate are treated as follows: the liquid digestate is sent to wastewater treatment, some of the solid digestate is incinerated, and some is sent to other specific treatments/uses. In addition, 14,021 t of high-quality compost is produced. The refuse from the composting process is incinerated or disposed of in landfills.

Once the definition of this scenario was agreed upon with the stakeholders [13], LCA and LCC analyses were performed to assess whether the alternative scenario was technically and economically feasible and environmentally desirable (i.e., that the alternative scenario performs better than the current scenario) [14,15].

2.5. Multi-Actor Identification of Operational (Non-Regulatory) Drivers and Barriers

Departing from the analysis of the regulatory drivers and barriers and having defined the alternative scenario, further operational (non-regulatory) drivers and barriers were identified, analyzed, and classified along with local stakeholders and international external experts in a dedicated participatory session.

Operational drivers and barriers were identified using the same definitions and classified under the same categories utilized in Section 2.2.

2.6. Multi-Actor Formulation of Policy Recommendations

The last step was the formulation of policy recommendations. Policy recommendations were related either to improving the potential enclosed in drivers or overcoming the barriers for the shift from the current to the alternative scenario. Where possible, policy recommendations were accompanied by GPs illustrating the proposal.

The initial formulation of policy recommendations was then independently revised and validated by the local stakeholders and international external experts to produce an improved version of the recommendations.

3. Results

This section includes the results from the identification of drivers and barriers plus the derived policy recommendations for a shift from the current to the alternative scenario. We focus on the two key stages of the value chain, namely the separate collection of municipal biowaste and the optimization of municipal biowaste treatment to maximize biomethane production and high-quality compost. For more details on identified drivers and barriers, please see the Supplementary Materials.

3.1. Drivers and Barriers to Improving the Separate Collection of Municipal Biowaste

According to the identified drivers and barriers, increasing the separate collection of municipal waste relies on a policy mix comprising the implementation of more efficient municipal biowaste separate collection schemes (i.e., DtD and smart bins) and financial incentives, including unit-based pricing (e.g., pay-as-you-throw, PAYT).

Regarding the implementation of high efficiency collection systems, the regulatory framework is well aligned at all administrative levels in terms of supporting the implementation of DtD and smart bin systems. The legislation explicitly includes the mandatory implementation of the separate collection of biowaste and the priority of high efficiency collection systems. However, based on previous experience in Catalonia, where the separate collection of biowaste has been intensely promoted during the last two decades, it is

uncommon that high efficiency collection systems are implemented unless it is supported by a broader kit of policy instruments and dedicated funding.

Stakeholders agreed on the relevance of the regulatory drivers in creating a favorable context for the implementation of high efficiency collection systems and pointed out several additional barriers. Public acceptance cannot be taken for granted, and the lack of public support is a relevant barrier, particularly when new collection systems are accompanied by an increase in waste charges. Previous experiences with the inconveniences of DtD collection systems during summer (e.g., hygiene-related issues) may also be seen as a barrier. Related to smart bins, the data processing and specialized skills required to run these models can be seen as a barrier to its implementation. Issues related to privacy and personal data protection were also raised. Also, high efficiency collection systems can be resource intensive when it comes to monitoring, and particularly when applying fines and sanctions. Also, the lack of environmental impact analyses comparing current and a more efficient biowaste collection system might hinder its implementation. Finally, large tourism flows in the MAB are a barrier to the implementation of high efficiency collection systems requiring user identification.

Landfill and incineration taxes are seen as drivers to improve the separate collection of biowaste. Catalonia has applied these taxes since 2004 and has regularly increased the tax rates to reinforce the incentive to separate collection. The tax rates applied in Catalonia, so far lower than in other European countries though significantly higher than in the rest of the Spanish regions, will soon be closer to European benchmarks, therefore exerting a more intense effect.

Binding targets related to separate collection rates and for increasing quality in terms of reducing impurities from the separate collection of biowaste are another complementary lever since high quality standards entail the implementation of high efficiency collection systems. In the case of the MAB, the lack of ambition of the current impurity targets included in the national regulatory framework can be seen as a barrier for obtaining high-quality compost as well as the fact that when more ambitious targets are set at the local level, no fines or sanctions are foreseen in case of non-accomplishment. Stakeholders pointed out that the fact that improved biowaste collection has a positive effect in terms of GHG emissions can be seen as a driver, since climate policy will support enhanced models for collection of biowaste.

Also related to the quality of biowaste collection, the fact that biowaste will need to be collected in compostable bags might contribute to reduce impurities since plastic bags are one of the main refuse materials found at biowaste treatment plants. In the absence of collection systems involving the individual identification of users, the extent of its implementation is subject to the good will of the users. Also, it should be considered that compostable bags made of bioplastics need to comply with EN 13,423 [29], suitable for industrial, but not necessarily home composting (different abiotic conditions). Furthermore, depending on the sorting process for input and output material, bioplastics in biowaste streams can pose a challenge to biowaste (pre-) treatment plants [30]. There is still a lack of education programs on proper separate collection of bioplastics and waste prevention, as some bioplastics (if littered) can also be a problem for aquatic and terrestrial ecosystems [31]. Furthermore, there is public confusion about the suitability of bioplastics for home composting as there is no common European standard (so far there are national standards such as the French standard NF T 51-800 [32]).

Unit-based pricing charges such a PAYT systems are considered a second pillar of an improved separate collection of biowaste. Furthermore, the regulatory framework set several drivers for a better funding of biowaste collection systems such as making full cost recovery mandatory. On the other hand, PAYT is explicitly recommended at all administrative levels. In practice, the implementation of PAYT and similar systems rely on user identification and, therefore, in the implementation of high efficiency collection systems.

Additionally, stakeholders identified several other social barriers such as the lack of knowledge on correct source separation of biowaste, particularly regarding bioplastics.

3.2. Policy Recommendations to Improving the Separate Collection of Municipal Biowaste

Based on the identified regulatory and non-regulatory drivers and barriers hindering the improvement of separate collection of municipal biowaste, a set of related policy recommendations were produced. First, even if already in place in Catalonia and, therefore, in the MAB, landfill and incineration taxes were acknowledged as a key policy instrument to support the implementation of high efficiency separate collection systems. In the absence of this instrument, the incentives for the implementation of any other instruments would be significantly weaker, as it can be observed in the rest of Spain, where separate collection rates are lower. Plus, the fact that the tax revenue from the landfill and incineration tax has been earmarked to improve waste management has allowed specific funding for separate collection and treatment of biowaste, which has boosted its implementation. Therefore, this lesson learnt can be seen as a policy recommendation and GP.

For landfill and incineration taxes to work as an incentive, tax rates should be set at a level that cost-efficiently contributes to accomplish waste management targets. In this sense, landfill and incineration tax rates in Catalonia were $65.3 \notin/t$ and $32.7 \notin/t$ in 2023, respectively, and are set to be increased up to $71.6 \notin/t$ and $35.8 \notin/t$ in 2024. Therefore, supported by the good performance of landfill and incineration taxes, a sustained increase in tax rates is recommended at the regional level after 2024.

A second policy recommendation at the national level was making the introduction of high efficiency collection systems (e.g., DtD, smart bin collection) mandatory in areas where the separate collection of biowaste has not been deployed so far or where results are poor. In parallel, it is recommended that the introduction of more open street container collection systems is prohibited. For those areas where the separate collection of biowaste is already carried out, it is advisable to introduce economic incentives for regional and local authorities for shifting to more efficient collection systems. This measure should be accompanied by appropriate funding and subsidies to ensure that the provisions from the Spanish Law 7/2022, where these systems are required to be prioritized, are made operational.

Two GPs support this recommendation: the cities of Münster (Germany) with a population of 317,713 inh., and Albano Laziale (Italy) with a population of 41,598 inh., introduced DtD collection with biowaste barcode bins more than 20 years ago. Results show that separate collection rates reached 87% and 94%, respectively, in 2019 [33]. There are also local examples from the MAB, in Corbera de Llobregat with 15,210 inhabitants, where biowaste collection rates rocketed from 23 kg/cap/year (impurities; 11.3%) to 84 kg/cap/year (impurities: 2.9%) after the implementation of DtD collection in February 2022.

A third set of recommendations involves regulating the use of PAYT to be mainstreamed as the baseline charging system, regarding municipal waste charges is a third central recommendation. The scope of the application of this measure, given the Spanish legal framework, relies on the national level. Charging systems based on the actual amount of waste generated create incentives for separating biowaste (and other recyclables) at source and reducing the amount of mixed residual waste. In Spain, according to Law 7/2022, local authorities shall introduce a charge or, where appropriate, a fee, which shall be specific, differentiated, and cover all waste management costs by 2025. It also states that waste charges should allow for the introduction of PAYT schemes. When applied together with well-designed collection of different waste fractions (including residual waste, biowaste, and recyclables), and adequately supported by communication campaigns, PAYT has been associated with an increase in the overall separate collection rates [34].

A fourth set of recommendations are related to the use of binding targets. For example, the introduction of binding targets regarding maximum biowaste content in mixed residual waste is requested to incentivize the proper collection and recycling of biowaste. According to Zero Waste Europe, such a cap could be set at 25 kg/cap/year by 2030 and 15 kg/cap/year by 2035. This recommendation was addressed at the EU level, although it could be suitable at the national and regional levels.

A fifth group of recommendations relates to information and awareness-raising. For example, at regional level, it is recommended to introduce sensibilization campaigns to improve the quality of source-separated biowaste. This can be performed with a traditional approach or through KAYT (know-as-you-throw), which means providing individuals with continuous, tailored feedback so they can correct their source separation behavior when necessary [35]. As a GP, the case of Bergamo in 2020 illustrates how KAYT works. The fact that people felt monitored was an incentive to change their behavior. In Bergamo, the concept of a 'neighborhood premium' will also be introduced as a collective economic incentive based on the analysis of bag collection data [36]. In general, the use of awareness raising campaigns about the relevance and technical details about the separate collection of biowaste was adopted as a recommendation to ease the transition towards the implementation of high efficiency collection systems. These campaigns should target citizens but also tourists. Raising awareness in the form of training for workers of the municipal waste collection services was also recommended to ensure sufficient knowledge of aspects related to regulations and targets, practical aspects of management, data collection and reporting, costs, and financing of the service.

A sixth recommendation focuses on large cities and provides them with targeted strategies and incentives for improving separate collection. The MAB contains several large cities and the implementation of high efficiency collection systems in these contexts differ from smaller urban and rural areas. A GP example for a larger city is Milan (Italy) with 1.3 million inhabitants, a density above 7000 inhabitants/km², and over 80% of the population living in high-rise buildings. In February 2012, the first step of introducing the separate collection of biowaste was to replace the traditional black bag with a transparent bag for the collection of the mixed residual waste, to facilitate the monitoring of the collection [37]. Then, the municipality introduced DtD collection. The yearly quantity of collected biowaste per capita rose from 28 kg/cap/y in 2011 to 110 kg/cap/y in 2019, almost 6 times higher than the EU average. The quality of separately collected biowaste was very high, with a share of only 4.3% of non-compostable materials in 2019, fully compatible with the subsequent treatment by the Italian Association for Composting and Biogas (CIC). Separate collection of municipal waste in general increased from 35% in 2011 to 63% in 2020, as other waste fractions are also targeted for DtD collection.

In MAB municipalities where DtD is already running, it is recommended to introduce clearly communicated sorting obligations for citizens and enforce the obligation by introducing controls on the separate collection of biowaste and the proportion of organic content in mixed residual waste. Where necessary, it is needed to make use of the corresponding fines for non-compliance (according to PREMET25). DtD collection alone does not necessarily lead to a higher biowaste collection quality and, therefore, support from inspections is needed. In the MAB, controls are carried out by the operator at the time of collecting the bin or container (in the case of commercial collection). In the event of non-compliance with the sorting obligations, the bin or container is left standing, and the non-compliance is reported with a sticker as a notice/warning. If the stickers do not lead to an improvement, an official letter can be sent, which may result in a fine. But so far, few municipalities have issued fines. In the case of smart bin collection, the use of approved and coded bags could be enforced, but this is more complicated and less effective. In the city of Münster, a four-phase "Biowaste bin campaign" ("Aktion Biotonne") was launched in 2017, an information campaign combined with a quality check of the biowaste bins of each household biowaste (DtD collection with barcode identification). If the quality found in the bins does not meet the standards, citizens receive a yellow card with instructions for better sorting. If the problem persists, the biowaste bin is removed and households must put the biowaste exclusively in the mixed residual waste bin, paying the highest annual fee. As a result, impurities decreased from 3.5% in mid-2017 to 1.9% in early-2018, indicating that the information campaign influenced separation behavior [33].

A final recommendation directed at the national level is to make yearly waste composition analyses for separately collected biowaste and mixed waste necessary. Knowing the composition of the separately collected biowaste provides information about the proportion and composition of impurities, and the composition of the mixed residual waste fraction provides information about the remaining recyclables, including organic waste. Both types of sorting analysis are crucial for improving separate (bio)waste collection and communication campaigns for the public. In the GP example of Milan, sorting analyses are carried out every 6 months. Knowing the exact composition of the mixed residual waste also helps to estimate the actual success of separate biowaste collection and biowaste prevention measures. In Milan's case, only 14% of food waste remains in the mixed residual waste [38].

3.3. Drivers and Barriers to Enhancing the Production of Biogas to Be Upgraded into Biomethane

In the MAB pilot area, biomethane obtained from AD of municipal biowaste is another key point of optimization towards increased circularity of the system. The drivers for enhancing the quality and quantity of separate collection go hand in hand with the drivers for increasing the number and capacity of AD plants and hence the production and use of biogas and biomethane. A clear environmental driver for biomethane is the reduction of GHG emissions through replacing fossil natural gas with biomethane.

Large-scale expansion also requires planning security for long-term investments, i.e., a stable regulatory framework, medium- to long-term security for subsidies, and support from local institutions. In Spain, as a main driver for the promotion of biogas and biomethane generation, the competent authorities shall promote the use of biogas from AD for energy purposes, for direct use in the plants themselves, as transport fuel, as feedstock for industrial processes and for injection into the natural gas grid in the form of biomethane, where technically and economically feasible [27]. At the EU level, various directives and national regulations support the development of renewable energy, including biomethane. A strong national driving force are, in part, the more ambitious targets than at the EU level, e.g., a 42% share of renewable energy in final energy consumption by 2030 [39] compared to the EU target of 32% renewable energy by 2030 [40].

An enhanced uptake of renewable hydrogen, biogas, and biomethane by industry is seen as a key contributor to the RePowerEU plan's objectives of diversifying gas supply and reducing the EU's dependence on Russian fossil fuels, while at the same time decreasing dependence on fluctuating natural gas prices and tackling the climate crisis [41]. In 2022, the European Commission published a Staff working document [18] which aims to increase biomethane production to 35 billion cubic meters (bcm) per year by 2030 and set the stage for further increasing the potential by 2050. This document contains several possible actions to unlock the potential of biogas and biomethane across all EU countries and explicitly promotes the use of secondary feedstock (crop residues, manure, organic waste), while discouraging the use of food and feed for biomethane production. The promoted multistakeholder involvement on defining the biomethane is a clear driver for local authorities to establish a specific framework for collaboration with the academic community and industry sector. In response to the REPowerEU Action Plan, the Spanish Biogas Roadmap was introduced in 2022 [42], setting a minimum biogas target of 10.41 TWh (1.07 bcm) per year by 2030. In addition, a biomethane target was introduced where, in 2030, at least 1% (0.39 bcm) of total gas consumption of the natural gas grid (2021: 377,157 GWh = 38.61 bcm) should be biomethane [43] However, in 2022, only 2% of total gas consumption in Spain was biogas [44]. As a barrier, both targets are rather unambitious and not in line with the RePowerEU target of 35 bcm for the EU by 2030 [18]. The Spanish biomethane production potential is estimated to be up to 20 bcm/y in 2050 [45], which would place Spain only behind France and Germany with 22 bcm/y, respectively [44]. A driver for the use of biomethane in the transport sector is the third Clean Mobility Package, published in May 2018, which includes the first ever CO_2 emission standards for heavy-duty vehicles. The focus on tailpipe emissions does not allow fair competition between the different clean mobility solutions [46].

Currently, there is insufficient infrastructure to produce biogas from biowaste from the food and agricultural sectors, and its conversion into biomethane. As a further barrier, Spain's gas grid infrastructure needs adjustments to accept biomethane after purification and investments for a network expansion, especially in remote areas. Not all AD plants (limited in number in Spain anyway) are located near a connection point for feeding biomethane into the local gas grid so far, so they currently only have the option of producing compressed natural gas (CNG). Another important point is that ensuring the quality of natural gas in the distribution networks is crucial both for the functioning of the system and for the safety of end consumers. A technical obstacle to the use of biomethane is that a certain degree of purity and quality is required for injection into the natural gas grid or for use in vehicles [47]. Achieving this standard consistently can be a technical challenge. Even though RED II can be seen as a driver for fostering the integration of biomethane into European gas grids and its cross-border trade by promoting joint projects and shared efforts [40], specific natural gas quality requirements, including methane content, may vary from among EU Member States, creating a fragmented market and making trade difficult.

The administrative burden for the construction of new biogas and biomethane plants was identified as another obstacle. When it comes to biogas production, a major barrier is that the number of Spanish biogas plants (146 [42] and 81 planned plants [48]) is significantly lower compared to other countries such as Germany (>10,000 plants in 2020) [49], France (1700 plants in 2022) [50], and Italy (approx. 2000 plants in 2020) [51], even though Spain is a country with such an important agri-food sector and hence potentially usable biowaste resources for AD. A lack of biogas plants also means that there are few biomethane upgrading plants. In October 2021, there were 1023 biomethane production plants in Europe, with Germany showing the highest energy production (>80,000 GWh [44]). Spain currently has only five operational biomethane production plants, reaching 95 GWh in 2020 (0.01 bcm [45]). The initial capital cost for the improvement of existing AD plants and the construction of new AD plants and biogas upgrading technologies is also a significant barrier. For the construction of new AD facilities, capacity building is needed in the municipalities to speed up the approval process, as human and financial resources are currently insufficient.

To enable an increase in biomethane production, investments in research and development of innovative technologies for the sustainable production of biogas and biomethane are required. In order to facilitate new investments in biogas and biomethane, the European Commission also offers extended access to grants and loans from existing EU funds, such as the Common Agricultural Policy Rural Development Funds, Structural and Cohesion Funds, National Resilience, and Recovery Plans, Horizon Europe, Innovation and Modernisation Funds, LIFE funds, and other national funds, as well as access to funds for innovative production and use of biomethane and biogas projects. Nevertheless, there is a lack of awareness of these grants as well as administrative challenges in applying for them, especially for SMEs and small municipalities.

Subsidy and support mechanisms, as well as a regulatory and fiscal frameworks that regulate the various applications of biogas, are fundamental driving forces. Without subsidies or a strong economic incentive, the cost of producing biomethane can be higher than that of natural gas or other energy sources. To attract investors, the business model for biomethane production needs to be guaranteed. Without a real and stable biomethane market, this is complicated.

The general public's increasing awareness and concern about climate change and environmental issues drive demand for cleaner, renewable energy sources, including biomethane. As a major driver of increasing the public awareness and facilitating the trading and recognition of biomethane in the energy market, in 2023, Spain established a Guarantee of Origin (GoO) system for renewable energy as a means to certify the renewable nature of 1 MWh of gas while listing the specifics of its production, i.e., when and where it was produced, the type of production facility, and energy source, etc. [52]. This trade mechanism is managed by the National Commission of Markets and Competition (CNMC) and is in line with European directives that encourage Member States to establish such certification mechanisms to promote transparency and consumer confidence in renewable energy.

3.4. Policy Recommendations to Introduce and Increase the Production of Biomethane

So far, there are no specific targets for biomethane production at local and regional levels, which is why the focus is on the EU and national context. For Spain, main policy recommendations include the introduction of a higher minimum biogas target (at least 5 bcm by 2030) and biomethane target (at least 5% of the total gas consumed) linked to the target for new plants to be built into the Spanish Biogas Roadmap [42].

At the EU level, the biomethane quality requirements for gas grid injection vary between Member States [47]. At the national level, technical support for the safe injection of biomethane into the gas grid, the expansion of the existing gas network, as well as the connection of decentralized generation plants with consumption centers over large distances (for example, dispersed settlements as specified at the EU level [18]) should be provided. At the national level, the reduction of the current delay in permitting procedures for new AD plants should be incentivized by increasing the capacity in municipalities responsible for issuing permits. In addition, GPs such as the establishment of one-stop shops for biomethane-related permits should be streamlined and a maximum processing time according to the EU biomethane targets [18] set. In addition, the upscaling of research pilot plants to an industrial scale should be facilitated.

At the EU and national levels, a 'well-to-wheel' approach should be emphasized, i.e., taking into account the whole fuel chain including extraction, production, transport, and use of the fuel or electricity [53], which would allow for a more realistic comparison of the different options, as proposed by the European Biogas Association (EBA) and Natural and bioGas Vehicle Association (NGVA), Europe [54]. Currently, Italy has the most methane refueling stations; 25% of all refueling stations in the EU [55]. A "well-to-wheel" study conducted in Denmark demonstrated that the use of compressed and liquefied natural gas results in a reduction of GHG emissions by 15–27% per kilometer driven compared to conventional fuels in both transport applications and for all vehicle classes. The effect is particularly large, 81–211%, when compressed and liquefied renewable natural gas is used alternatively. The outcomes depend on the type and source of feedstock used, the vehicle engine, the assumed methane leakage and slip, and the allocated energy and environmental credits for digestate in each pathway [56]. At the EU level, there is a target of 14% renewables in the transport sector by 2030 [40]. In 2021, the share of renewable energy in the Spanish transport sector was around 9% of the EU average compared to Sweden with 30.4% [57].

At the national level, it is crucial to introduce financial incentives for bioenergy compared to fossil energy through measures such as a minimum purchase price or incentive tariffs for biomethane with a duration of 15 years (e.g., as enshrined in Italian national law [58]). Another option is negative incentives for the use of fossil energy (e.g., tax increases). At the EU level, an incentive mechanism based on the assessment of the carbon footprint would also be conceivable, i.e., the lower the carbon footprint $[CO_2 g/MJ]$, the higher the price/incentive should be for this product/energy source (instead of having a flat incentive rate). The carbon footprint assessment should be performed according to the International Sustainability and Carbon Certification (ISCC) [59]. An exclusive dedication of the landfill/incineration taxes for the improvement of the current biowaste management including the construction of new AD facilities is recommended.

At the EU level, Public Private Partnerships (PPP) for bio-based industries, e.g., the Circular Bio-based Europe Joint Undertaking (CBE JU) [60], should be strengthened. In addition, the Spanish participation in the Biomethane Industrial Partnership [61] should be encouraged. The implementation of innovative technologies for the sustainable production of biogas and biomethane based on the gasification of biogenic waste as specified at the EU level [18] needs support at the national level. Advances in purification technologies, like pressure swing adsorption or membrane technologies [47], have made it more efficient and economical to upgrade biogas to biomethane suitable for grid injection or vehicular use. Ongoing research and development also include technologies such as "biomethanation", which is converting the CO_2 fraction from the biogas into CH_4 together with added hydro-

gen (H₂) using excess electricity, producing the same amount of biomethane as the amount of biogas used [62]. However, this technology is dependent on the price of hydrogen. The use of a power-to-gas technology, i.e., converting the surplus electricity from solar or wind energy into hydrogen and convert this hydrogen (together with CO₂) into biomethane [63], might be a possibility and should be examined in the specific local context. In general, the dissemination/utilization of research results needs to be facilitated, e.g., from the SEMPRE-BIO project [64], in which three different technologies for the production of biomethane with three different raw materials are being tested, or from the LIFE INFUSION project [65], in which biogas, nutrients, and treated water are obtained from landfill leachate and liquid digestate from organic municipal waste.

National awareness campaigns on the recently introduced Guarantee of Origin (GoO) system, which consists of demonstrating to an end customer that a certain percentage or amount of energy has been generated from renewable sources [52], are recommended to facilitate the introduction of biomethane in the market and increase its demand.

4. Discussion

The key role of municipal biowaste as feedstock in CBE strategies has been underlined in this work. The availability of biowaste as a raw material is, nevertheless, closely linked to measures to improve the quality and quantity of separately collected municipal biowaste [66]. The current European collection rates of biowaste compared to their total generation (average 34%) show that separate collection is still in its infancy. Therefore, the implementation of biowaste (especially food waste) collection policies and practices is a must in order to increase overall recycling rates in the near future [1]. High efficiency separate collection systems are the basic building blocks for the successful implementation of other measures in the rest of the value chain, i.e., implementing the technology required for biomethane upgrading is pointless unless high efficiency collection systems are in place in advance. Plus, these systems do not produce sufficient shifts in citizen's behavior and should be accompanied by a mix of other instruments including economic incentives [34] and communication campaigns. Open street bins are obsolete when it comes to achieving significant collection rates, because these systems do not allow for user identification (e.g., such as DtD or smart bin collection) and, therefore, the range of incentives applicable and its effectiveness is limited.

The point about the effectiveness of high efficiency collection systems and related incentives as compared to street open bins was already well known by all stakeholders and policy-makers during the local Living Labs. Furthermore, there is enough scientific and practical evidence and knowledge about its effectiveness from decades ago, including case studies from Catalonia [67–69]. However, a still pending barrier that prevents further changes from being implemented is related to governance and the fact that 'command-and-control' measures require political leadership and consensus along with public funding.

Securing high quality biowaste supply is crucial for biomethane projects. Indeed, largescale biomethane production, especially if based on energy crops, may raise concerns about land use change, biodiversity loss, and competition with food crops [70,71]. While Spain has abundant agricultural residues, there can be competition for these feedstocks for other purposes such as composting, animal bedding, or as raw materials for other industries [72]. For these reasons, a focus on municipal biowaste as a consistent and economical feedstock for biomethane plants is even more important.

In addition to a continuous higher quality and quantity of biowaste, new AD plants are also required to produce biomethane from biogas. The construction of new AD plants, the possible capacity increase of the existing ones or the adaptation of the existing Mechanical Biological Treatment (MBT) to AD treating only source-separated biowaste, require a stable regulatory framework that ensures viability and security, particularly in terms of energy production, so that it is competitive with traditional fossil fuels [73]. As far as financial support for the construction of new AD plants is concerned, it should be noted that the effectiveness of state subsidies also depends on the size of the plant. A case study in southern Spain illustrates that plants of $150 \text{ m}^3/\text{h}$ (medium scale) are profitable if 10% of the investment is subsidized, while smaller plants do not reach profitability even if 50% of the investment is subsidized [74]. What also should be considered is the usage of BAT in order to maximize the methane yield, to save energy, and to minimize environmental impacts [66,75].

Using biomethane derived from the AD of source-separated biowaste in gas networks could support the decarbonization of the energy sector and the transition to circular economy models and more sustainable cities, reduce energy dependency, and generate new employment opportunities and markets [76,77]. To foster the growth of the biomethane sector in the MAB, the current barriers need to be addressed through a combination of regulatory support, technological advancements, public awareness campaigns, and economic incentives. Even though "Ruta de Biogas" [42] is an important document, Spain is still lacking comprehensive and supportive regulatory framework tailored for biomethane production, injection, and usage. Targeted policy measures and incentives to support and establish a stable medium and long-term biomethane market in Spain and in countries with similar characteristics are needed [73].

Biomethane has also the potential to replace natural gas as a sustainable vehicle fuel, but governmental support is needed [78]. The reduced external costs of biomethane-powered vehicles compared to petrol-powered vehicles could serve as a basis for policy decisions to encourage the growth of biomethane use in the transport sector, leading to a reduction in GHG emissions [79]. Incentivizing the local production of biomethane supports the potential transition to a more sustainable waste transport system in the MAB, using biomethane instead of natural gas as a fuel for the waste collection fleet and hence reducing GHG emissions. In addition, the substitution of fossil fertilizers with high-quality compost could further contribute to decarbonization, lower consumption of raw materials, and less pollution [80].

Overall, the economic factors that can have the greatest impact on the chain's performance are waste disposal fees, the selling price of biomethane, and economic incentives for biomethane producers. However, it is important to rebalance the economic and environmental performance [81]. Among market-based instruments, feed-in support schemes and quota obligations for biomethane are the most important drivers in Europe. To allow the biomethane sector to compete with fossil fuels, policy should focus on reducing existing support mechanisms for electricity producers on the one hand and subsidizing biomethane production on the other. Tax policies aimed at increasing the price of competing fossil fuels would bring additional benefits [82].

The allocation of subsidies is seen as an essential prerequisite for promoting the biomethane transport sector. Subsidies should be combined with other policy measures, supporting a significant change in infrastructure, involving the provision of compressed natural gas (bio-CNG) refueling stations and biomethane-powered vehicles [83]. In the beginning (as with other successful renewable energy schemes) greater incentives would be needed to get the industry off the ground, but these subsidies could be phased out over time [76]. Member States have adopted national policy frameworks to foster alternative fuels in the EU, setting individual targets for the period 2020–2030, leading to a possible fragmentation of alternative fuel infrastructure [84], which needs to be harmonized in the near future. At the EU level, carbon footprint-based incentives for biomethane (avoided carbon footprint) should be linked to the emission commitments of industry or other sectors, i.e., those subject to the EU Emissions Trading System for GHG (EU ETS [85]).

5. Conclusions

In summary, in the course of this study, the two main research questions could be answered: The main drivers and barriers for the implementation of the selected alternative value chain in line with CBE principles were identified, i.e., 34 drivers and 18 barriers detected in the policy framework through desk research and 14 drivers, and 19 barriers determined through stakeholder engagement. The drivers and barriers, together with examples of GPs, were validated by relevant stakeholders and translated into a set of policy recommendations.

Policies to improve the performance of a local biowaste management system should recognize the central role of different stakeholders, whose site-specific knowledge must play a key role in decision-making. Therefore, the identified policy recommendations can accelerate the transition towards a more circular biowaste management in the MAB. To our knowledge, this is the first comprehensive, multi-actor study in Spain that cocreates knowledge about the drivers, barriers, and policy recommendations to improve the quantity and quality of source-separated biowaste, with the aim of promoting biogas production from AD and upgrading it into biomethane.

These results are relevant to European/international audiences given (1) the recently introduced obligation for all European cities to separate biowaste through biowaste collection systems (in Spain: with priority for DtD or smart containers as analyzed in the alternative scenario); and (2) the commitment of EU countries to be in line with the new RePowerEU plan and the need to increase the biomethane production.

Despite the local focus of the analysis, the list of policy recommendations can be used as a starting point to be tailored to the context of other EU cities, of which most are still far from complying with CBE principles as they struggle to reconcile legislation, waste management, circular economy, and bioeconomy.

Further research should be conducted in other urban areas and regions in Spain and beyond, where the circular bioeconomy potential of municipal biowaste has not yet been fully exploited.

Supplementary Materials: The following supporting information can be downloaded at: https://www.mdpi.com/article/10.3390/su16031208/s1; General Information about anaerobic digestion and biomethane; Table on regulatory drivers and barriers to improving the separate collection of municipal biowaste and production and use of biomethane in the MAB; Table on non-regulatory drivers and barriers based on the feedback from stakeholders' involvement (MA approach) to improving the separate collection of municipal biowaste and production and use of biomethane in the MAB. Refs [86–101] are citied in Supplementary Materials.

Author Contributions: Formal analysis, writing—original draft preparation, project administration, K.M.; formal analysis, writing—original draft preparation, visualization, S.S.; writing—review and editing, project administration, supervision, funding acquisition, writing—review and editing, supervision, I.P.-V.; writing—review and editing, project administration, supervision, funding acquisition, R.C.; investigation, formal analysis, data curation; writing—review and editing, L.M.S.; investigation, data curation, L.C.; formal analysis, data curation, G.F.; formal analysis, data curation; writing—review and editing, A.Z. All authors have read and agreed to the published version of the manuscript.

Funding: The work described in this paper was carried out within the BIOCIRCULARCITIES (BCC) project, which ran from October 2021 to September 2023. This project has received funding from the Bio-based Industries Joint Undertaking (JU) under the European Union's Horizon 2020 research and innovation programme under Grant Agreement No. 101023516. The JU receives support from the European Union's Horizon 2020 research and innovation programme and the Bio-based Industries Consortium.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data is contained within the article and Supplementary Materials.

Acknowledgments: The authors would like to acknowledge all the stakeholders engaged in the MAB local Living Labs for the precious discussions and feedback. Many thanks to Victor Mitjans from AMB (Àrea Metropolitana de Barcelona) for providing essential data and information for the performed LCA and LCC analyses on the current situation and alternative scenario. Special thanks also to the participants of our peer-review session Jimmy Roussel (LIST, Luxembourg Institute of Science and Technology), Andres Saldivia (Hysytech S.r.I.), and Michele Giavini (ARS Ambiente Srl),

as well as our advisory board members John Vos (BTG Biomass Technology Group BV) and Francesc Giró i Fontanals (ARC, Catalan Waste Agency). Special thanks also to Mike Stinavage (Fundació ENT) for the language revision of the text.

Conflicts of Interest: The authors declare no conflicts of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

References

- 1. Favoino, E.; Giavini, M.; Rupp, M. *Bio-Waste Generation in the EU: Current Capture Levels and Future Potential*; This Report Was Commissioned by the Bio-based Industries Consortium (BIC); Zero Waste Europe: Brussels, Belgium, 2020.
- The European Parliament and the European Council. Directive 2008/98/EC Directive 2008/98/EC of the European Parliament and of the Council of 19 November 2008 on Waste and Repealing Certain Directives (Waste Framework Directive). Off. J. Eur. Union 2008, L312, 1–59.
- 3. COM/2020/98 Final Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: A New Circular Economy Action Plan for a Cleaner and More Competitive Europe; European Commission: Brussels, Belgium, 2020.
- 4. COM/2018/673 Final Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions—A Sustainable Bioeconomy for Europe: Strengthening the Connection between Economy, Society, and the Environment; European Commission: Brussels, Belgium, 2018; Volume 10.
- 5. European Commission. *Bioeconomy: The European Way to Use Our Natural Resources;* Action Plan 2018; European Commission: Brussels, Belgium, 2018.
- 6. COM/2019/640 Final COM/2019/640 Final. *Communication from the Commission to the European, the European Council, the Council, the European Economic and Social Committee and the Committee of the Regions. The European Green Deal;* European Commission: Brussels, Belgium, 2019; pp. 1–24.
- D'Amato, D.; Droste, N.; Allen, B.; Kettunen, M.; Lähtinen, K.; Korhonen, J.; Leskinen, P.; Matthies, B.D.; Toppinen, A. Green, Circular, Bioeconomy: A Comparative Analysis of Sustainability Avenues. J. Clean. Prod. 2017, 168, 716–734. [CrossRef]
- 8. Lieder, M.; Rashid, A. Towards Circular Economy Implementation: A Comprehensive Review in Context of Manufacturing Industry. *J. Clean. Prod.* 2016, *115*, 36–51. [CrossRef]
- 9. Kardung, M.; Cingiz, K.; Costenoble, O.; Delahaye, R.; Heijman, W.; Lovrić, M.; van Leeuwen, M.; M'barek, R.; van Meijl, H.; Piotrowski, S.; et al. Development of the Circular Bioeconomy: Drivers and Indicators. *Sustainability* **2021**, *13*, 413. [CrossRef]
- European Commission; Directorate-General for Research and Innovation; Platt, R.; Bauen, A.; Reumerman, P.; Geier, C.; Van Ree, R.; Vural Gursel, I.; Garcia, L.; Behrens, M.; et al. EU Biorefinery Outlook to 2030: Studies on Support to Research and Innovation Policy in the Area of Bio-Based Products and Services; Publications Office: Brussels, Belgium, 2021. [CrossRef]
- 11. Zuin, V.G.; Ramin, L.Z. Green and Sustainable Separation of Natural Products from Agro-Industrial Waste: Challenges, Potentialities, and Perspectives on Emerging Approaches. *Top. Curr. Chem.* **2018**, *376*, 3.
- 12. Lokesh, K.; Ladu, L.; Summerton, L. Bridging the Gaps for a "circular" Bioeconomy: Selection Criteria, Bio-Based Value Chain and Stakeholder Mapping. *Sustainability* **2018**, *10*, 1695. [CrossRef]
- 13. Fiorentino, G.; Ansanelli, G.; Ncube, A.; Zucaro, A. *State of the Art of Biowaste Production and Management in the Pilot Areas*; Deliverable D2.1 of WP2; Biocircularcities Project: Brussels, Belgium, 2022.
- 14. Ansanelli, G.; Fiorentino, G.; Zucaro, A. LCA and LCC Analyses of the Selected Systems Producing and Managing Biowaste in the Pilot Areas; Deliverable D2.2 of WP2; Biocircularcities Project: Brussels, Belgium, 2023.
- 15. Ansanelli, G.; Fiorentino, G.; Zucaro, A. LCT Recommendations through the LCA and LCC Analyses of the Selected Systems Producing and Managing Biowaste in the Pilot Areas; Deliverable D2.3 of WP2; Biocircularcities Project: Brussels, Belgium, 2023.
- 16. Meisterl, K.; Chifari, R. *Regulatory Gap and Opportunity Analysis for a Circular Bioeconomy*; Deliverable D3.2 of WP3; Biocircularcities Project: Brussels, Belgium, 2023.
- 17. Meisterl, K.; Chifari, R. *Policy Recommendations for Implementing Circular Bioeconomy in the Pilot Areas*; Deliverable D3.3 of WP3; Biocircularcities Project: Brussels, Belgium, 2023.
- SWD/2022/230 Final SWD/2022/230final: Commission Staff Working Document: Implementing the Repower EU Action Plan: Investment, Hydrogen Accelerator and Achieving the Biomethane Targets, Accompanying COM/2022/230 Final; European Union: Brussels, Belgium, 2022.
- Rogelj, J.; Den Elzen, M.; Höhne, N.; Fransen, T.; Fekete, H.; Winkler, H.; Schaeffer, R.; Sha, F.; Riahi, K.; Meinshausen, M. Paris Agreement Climate Proposals Need a Boost to Keep Warming Well below 2 °C. *Nature* 2016, 534, 631–639. [CrossRef] [PubMed]
- 20. Weiland, P. Biogas Production: Current State and Perspectives. Appl. Microbiol. Biotechnol. 2010, 85, 849–860. [CrossRef]
- 21. Salemdeeb, R.; Saint, R.; Clark, W.; Lenaghan, M.; Pratt, K.; Millar, F. A Pragmatic and Industry-Oriented Framework for Data Quality Assessment of Environmental Footprint Tools. *Resour. Environ. Sustain.* **2021**, *3*, 100019. [CrossRef]
- 22. Kambanou, M.L.; Sakao, T. Using Life Cycle Costing (LCC) to Select Circular Measures: A Discussion and Practical Approach. *Resour. Conserv. Recycl.* 2020, 155, 104650. [CrossRef]

- 23. Schütz, F.; Heidingsfelder, M.L.; Schraudner, M. Co-Shaping the Future in Quadruple Helix Innovation Systems: Uncovering Public Preferences toward Participatory Research and Innovation. *J. Des. Econ. Innov.* **2019**, *5*, 128–146. [CrossRef]
- 24. Chifari, R.; Lo Piano, S.; Bukkens, S.G.F.; Giampietro, M. A Holistic Framework for the Integrated Assessment of Urban Waste Management Systems. *Ecol. Indic.* 2018, 94, 24–36. [CrossRef]
- Lopez, M.; Chifari, R. Policy Framework and Good Practices on Circular Bioeconomy and Biowaste Management; Deliverable D3.1 of WP3; Biocircularcities Project: Brussels, Belgium, 2021.
- AMB Àrea Metropolitana de Barcelona. Population 2021. Available online: https://www.amb.cat/s/es/web/areametropolitana/coneixer-l-area-metropolitana/poblacio.html (accessed on 6 November 2023).
- Law 7/2022 National Law 7/2022 on Waste and Contaminated Soils for a Circular Economy. Ley 7/2022, de 8 de Abril, de Residuos y Suelos Contaminados para Una Economía Circular; Agencia Estatal Boletín Oficial del Estado: Madrid, Spain, 2022. Available online: https://www.boe.es/buscar/act.php?id=BOE-A-2022-5809 (accessed on 6 November 2023).
- 28. DECISIVE: A Decentralised Management Scheme for Innovative Valorisation of Urban Biowaste. Available online: https://www.decisive2020.eu/ (accessed on 10 December 2023).
- 29. European Bioplastics. EN 13432 Certified Bioplastics: Performance in Industrial Composting; European Bioplastics e.V.: Berlin, Germany, 2015.
- Hopewell, J.; Dvorak, R.; Kosior, E. Plastics Recycling: Challenges and Opportunities. *Philos. Trans. R. Soc. Lond. B Biol. Sci.* 2009, 364, 2115–2126. [CrossRef]
- 31. Accinelli, C.; Saccà, M.L.; Mencarelli, M.; Vicari, A. Deterioration of Bioplastic Carrier Bags in the Environment and Assessment of a New Recycling Alternative. *Chemosphere* **2012**, *89*, 136–143. [CrossRef]
- 32. NF T51-800; Plastics—Specifications for Plastics Suitable for Domestic Composting. Plastiques—Spécifications Pour Les Plastiques Aptes Au Compostage Domestique NF T51-800. C-Label: Istanbul, Turkey, 2015.
- 33. Hoop; Valuewaste; Scalibur; WaysTUP!; Cityloops. *ROOTS-Circular Policies for Changing the Biowaste System Position Paper;* European Biomass Industry Association: Brussels, Belgium, 2022.
- Morlok, J.; Schoenberger, H.; Styles, D.; Galvez-Martos, J.-L.; Zeschmar-Lahl, B. The Impact of Pay-As-You-Throw Schemes on Municipal Solid Waste Management: The Exemplar Case of the County of Aschaffenburg, Germany. *Resources* 2017, 6, 8. [CrossRef]
- Font Vivanco, D.; Bassa Echaurren, M.; Aulinas Masó, M.; Giavini, M.; Doubell, M.; Bagatella, E.; Venturi, R. Carbon Footprint and Life Cycle Economic Costs of Pilot Know-as-You-Throw Schemes in Italy and Spain. Available online: https://ssrn.com/ abstract=4352966 (accessed on 12 November 2023).
- 36. Lisai, S.; Marengo, P.; Velache, C. Collection of Experiences about Pay as You Throw (PAYT) and Know as You Throw (KAYT); REthinkWASTE Project: Brussels, Belgium, 2021.
- Condamine, P. The Story of Milan. Successfully Collecting Food Waste for over 1.4 Million Inhabitants. Zero Waste Europe. 2012. Available online: https://zerowastecities.eu/wp-content/uploads/2021/11/Milan-Case-Study-1.pdf (accessed on 8 November 2023).
- Ricci-Jürgensen, M. Food Waste Collection in Metropolitan Areas: Milan (Italy). Presentation to the Environment Committee Meeting of the City of London. 2014. Available online: https://meetings.london.gov.uk/documents/b10746/Minutes%20 -%20Appendix%202%20-%20Food%20Waste%20in%20Milan%20Wednesday%2009-Jul-2014%2014.00%20Environment%20 Committee.pdf?T=9 (accessed on 8 November 2023).
- PNIEC Spain's National Integrated Energy and Climate Plan. Plan Nacional Integrado de Energía y Clima (PNIEC) 2021–2030; Ministerio para la Transición Ecológica y el Reto Demográfico: Madrid, Spain, 2020. Available online: https://www.miteco.gob. es/content/dam/miteco/images/es/pnieccompleto_tcm30-508410.pdf (accessed on 8 November 2023).
- 40. Directive (EU) 2018/2001 of the European Parliament and of the Council of 11 December 2018 on the Promotion of the Use of Energy from Renewable Sources. (RED II); European Commission: Brussels, Belgium, 2018.
- 41. COM/2022/230 Final: Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee and the Committee of the Regions—REPowerEU Plan; European Commission: Brussels, Belgium, 2022.
- Spanish Biogas Roadmap. Hoja de Ruta Del Biogás; Ministerio para la Transición Ecológica y el Reto Demográfico: Madrid, Spain, 2022. Available online: https://www.miteco.gob.es/content/dam/miteco/es/energia/files-1/es-es/Novedades/Documents/00HR_Biogas_V6.pdf (accessed on 8 November 2023).
- 43. Statistica. Spanish Natural Gas Consumption 2010–2021. 2021. Available online: https://es.statista.com/estadisticas/797237 /consumo-de-gas-natural-en-espana/ (accessed on 9 November 2023).
- 44. EBA Statistical Report 2022. Tracking Biogas and Biomethane Deployment across Europe; European Biomass Industry Association: Brussels, Belgium, 2022.
- 45. Guevara, P.R. Spanish Biomethane Here We Go! Análisis de las Potencias Europeas en Biometano para Entender por qué y cómo Replicar el Modelo en España. *Retema Rev. Técnica Medio Ambiente. Rev. Digit. Espec. Bioenergía* **2022**, 241, 60–72.
- 46. IRU Mobility Package 03—The Last Part of the European Commission's Mobility Package Proposals Covering Legislative Initiatives in the Areas of Safe, Clean and Connected Mobility; IRU: Geneva, Switzerland, 2018.
- Ryckebosch, E.; Drouillon, M.; Vervaeren, H. Techniques for Transformation of Biogas to Biomethane. *Biomass Bioenergy* 2011, 35, 1633–1645. [CrossRef]

- 48. Residuos Profesional MITECO Awards 76.5 Million to Unique Biogas Facility Projects. 2023. Available online: https://www.residuosprofesional.com/miteco-otorga-ayudas-biogas/ (accessed on 7 November 2023).
- Gustafsson, M.; Ammenberg, J. IEA Bioenergy Task 37-A Perspective on the State of the Biogas Industry from Selected Member Countries; IEA Bioenergy: Paris, France, 2022. Available online: https://task37.ieabioenergy.com/wp-content/uploads/sites/32/2022/02 /IEA_T37_CountryReportSummary_2021.pdf (accessed on 7 November 2023).
- 50. MéthaFrance In Numbers. Key Data on the Methanisation Sector in France and Associated Resources for Further Information. 2023. Available online: https://www.methafrance.fr/en-chiffres (accessed on 7 November 2023).
- 51. Energia Oltre. All the Numbers of Biomethane in Italy and Europe. 2023. Available online: https://energiaoltre.it/tutti-i-numeridel-biometano-in-italia-e-in-europa/ (accessed on 7 November 2023).
- 52. Order TED/1026/2022 of 28 October, Approving the Procedure for the Management of the System of Guarantees of Origin of Gas from Renewable Sources. Orden TED/1026/2022, de 28 de Octubre, por la que se aprueba el Procedimiento de Gestión del Sistema de Garantías de Origen del Gas Procedente de Fuentes Renovables; Ministerio para la Transición Ecológica y el Reto Demográfico 2022. Available online: https://www.boe.es/eli/es/o/2022/10/28/ted1026 (accessed on 7 November 2023).
- 53. Ferreira, S.F.; Buller, L.S.; Berni, M.; Forster-Carneiro, T. Environmental Impact Assessment of End-Uses of Biomethane. J. Clean. Prod. 2019, 230, 613–621. [CrossRef]
- EBA News. Overview on Key EU Policies for the Biogas Sector. 2019. Available online: https://www.europeanbiogas.eu/ overview-on-key-eu-policies-for-the-biogas-sector/ (accessed on 8 November 2023).
- 55. Tratzi, P.; Torre, M.; Paolini, V.; Tomassetti, L.; Montiroli, C.; Manzo, E.; Petracchini, F. Liquefied Biomethane for Heavy-Duty Transport in Italy: A Well-to-Wheels Approach. *Transp. Res. D Transp. Environ.* **2022**, *107*, 103288. [CrossRef]
- 56. Hagos, D.A.; Ahlgren, E.O. Well-to-Wheel Assessment of Natural Gas Vehicles and Their Fuel Supply Infrastructures—Perspectives on Gas in Transport in Denmark. *Transp. Res. D Transp. Environ.* **2018**, *65*, 14–35. [CrossRef]
- 57. Eurostat. Share of Renewables in Transport Decreased in 2021. 2023. Available online: https://ec.europa.eu/eurostat/web/ products-eurostat-news/w/ddn-20230123-2 (accessed on 8 November 2023).
- 58. Italian National Decree n. 240 of 15 September 2022. Development of Biomethane, According to Criteria to Promote the Circular Economy—Biomethane Production. DECRETO 15 Settembre 2022 Attuazione Degli Articoli 11, Comma 1 e 14, Comma 1, Lettera b), del Decreto Legislativo 8 Novembre 2021, n. 199, al Fine di Sostenere la Produzione di Biometano Immesso Nella Rete del Gas Naturale, in Coerenza con la Missione 2, Componente 2, Investimento 1.4, del PNRR. (22A06066). 2022. Available online: https://www.gazzettaufficiale.it/eli/id/2022/10/26/22A06066/sg (accessed on 8 November 2023).
- 59. ISCC International Sustainability and Carbon Certification. Available online: https://www.iscc-system.org/ (accessed on 8 November 2023).
- CBE JU. Circular Bio-Based Europe Joint Undertaking. Available online: https://european-union.europa.eu/institutions-lawbudget/institutions-and-bodies/search-all-eu-institutions-and-bodies/circular-bio-based-europe-joint-undertaking-cbe-ju_ en (accessed on 8 November 2023).
- 61. Biomethane Industrial Partnership. The Biomethane Industrial Partnership: Teaming up to Achieve 35 Bcm of Sustainable Biomethane by 2030. Available online: https://bip-europe.eu/ (accessed on 8 November 2023).
- 62. Angelidaki, I.; Treu, L.; Tsapekos, P.; Luo, G.; Campanaro, S.; Wenzel, H.; Kougias, P.G. Biogas Upgrading and Utilization: Current Status and Perspectives. *Biotechnol. Adv.* 2018, *36*, 452–466. [CrossRef] [PubMed]
- 63. Zábranská, J.; Pokorna, D. Research Review Paper Bioconversion of Carbon Dioxide to Methane Using Hydrogen and Hydrogenotrophic Methanogens. *Biotechnol. Adv.* 2018, *36*, 707–720. [CrossRef]
- 64. SEMPREBIO: Making Biomethane Production Efficient. Available online: https://sempre-bio.com/ (accessed on 7 November 2023).
- 65. LIFE INFUSION: Intensive Treatment of Waste Effluents and Conversion into Useful Sustainable Outputs: Biogas, Nutrients and Water. Available online: https://lifeinfusion.eu/ (accessed on 7 November 2023).
- 66. Wang, Y.; Yuan, Z.; Tang, Y. Enhancing Food Security and Environmental Sustainability: A Critical Review of Food Loss and Waste Management. *Resour. Environ. Sustain.* **2021**, *4*, 100023. [CrossRef]
- 67. Autoridad Independiente de Responsabilidad Fiscal (AIReF) Estudio Gestión de Los Residuos Municipales; Autoridad Independiente de Responsabilidad Fiscal (AIReF): Madrid, Spain, 2023.
- 68. Alvarez, M.D.; Sans, R.; Garrido, N.; Torres, A. Factors That Affect the Quality of the Bio-Waste Fraction of Selectively Collected Solid Waste in Catalonia. *Waste Manag.* 2008, 28, 359–366. [CrossRef] [PubMed]
- 69. Saldivia-Gonzatti, L.I.; Jannes, G.; Barreal, J. Factors Influencing the Rate of Sorted Solid Waste Collection: An Empirical Analysis towards Local Management in Catalonia (NE Spain). *Cities* **2022**, *131*, 104038. [CrossRef]
- Tamburini, E.; Gaglio, M.; Castaldelli, G.; Fano, E.A. Is Bioenergy Truly Sustainable When Land-Use-Change (LUC) Emissions Are Accounted for? The Case-Study of Biogas from Agricultural Biomass in Emilia-Romagna Region, Italy. Sustainability 2020, 12, 3260. [CrossRef]
- 71. Scarlat, N.; Dallemand, J.F.; Fahl, F. Biogas: Developments and Perspectives in Europe. *Renew. Energy* 2018, 129, 457–472. [CrossRef]
- 72. Searle, S.; Malins, C. White Paper: Availability of Cellulosic Residues and Wastes in the EU; International Council on Clean Transportation (ICCT): Beijing, China; Berlin, Germany; Brussels, Belgium; San Francisco, CA, USA; Washington, DC, USA, 2013.
- 73. Fernández-González, J.M.; Martín-Pascual, J.; Zamorano, M. Biomethane Injection into Natural Gas Network vs Composting and Biogas Production for Electricity in Spain: An Analysis of Key Decision Factors. *Sustain. Cities Soc.* 2020, *60*, 102242. [CrossRef]

- 74. Baena-Moreno, F.M.; Malico, I.; Rodríguez-Galán, M.; Serrano, A.; Fermoso, F.G.; Navarrete, B. The Importance of Governmental Incentives for Small Biomethane Plants in South Spain. *Energy* **2020**, *206*, 118158. [CrossRef]
- Commission Implementing Decision (EU) 2018/1147 of 10 August 2018 Establishing Best Available Techniques (BAT) Conclusions for Waste Treatment, under Directive 2010/75/EU of the European Parliament and of the Council (Notified under Document C (2018) 5070). Off. J. Eur. Union 2018, L 208, 38–90.
- 76. Rajendran, K.; O'Gallachoir, B.; Murphy, J.D. The Combined Role of Policy and Incentives in Promoting Cost Efficient Decarbonisation of Energy: A Case Study for Biomethane. J. Clean. Prod. 2019, 219, 278–290. [CrossRef]
- 77. D'Adamo, I.; Ribichini, M.; Tsagarakis, K.P. Biomethane as an Energy Resource for Achieving Sustainable Production: Economic Assessments and Policy Implications. *Sustain. Prod. Consum.* **2023**, *35*, 13–27. [CrossRef]
- 78. Adamo, I.D.; Falcone, P.M.; Gastaldi, M.; Morone, P. RES-T Trajectories and an Integrated SWOT-AHP Analysis for Biomethane. Policy Implications to Support a Green Revolution in European Transport. *Energy Policy* **2020**, *138*, 111220. [CrossRef]
- 79. Uusitalo, V.; Havukainen, J.; Soukka, R.; Väisänen, S.; Havukainen, M.; Luoranen, M. Systematic Approach for Recognizing Limiting Factors for Growth of Biomethane Use in Transportation Sector—A Case Study in Finland. *Renew. Energy* **2015**, *80*, 479–488. [CrossRef]
- 80. Gustafsson, M.; Svensson, N.; Eklund, M.; Dahl Öberg, J.; Vehabovic, A. Well-to-Wheel Greenhouse Gas Emissions of Heavy-Duty Transports: Influence of Electricity Carbon Intensity. *Transp. Res. D Transp. Environ.* **2021**, *93*, 102757. [CrossRef]
- 81. Fraccascia, L.; Spagnoli, M.; Riccini, L.; Nastasi, A. Designing the Biomethane Production Chain from Urban Wastes at the Regional Level: An Application to the Rome Metropolitan Area. *J. Environ. Manag.* **2021**, 297, 113328. [CrossRef]
- Barisa, A.; Kirsanovs, V.; Safronova, A. Future Transport Policy Designs for Biomethane Promotion: A System Dynamics Model. J. Environ. Manag. 2020, 269, 110842. [CrossRef] [PubMed]
- 83. D'Adamo, I.; Falcone, P.M.; Huisingh, D.; Morone, P. A Circular Economy Model Based on Biomethane: What Are the Opportunities for the Municipality of Rome and Beyond? *Renew. Energy* **2021**, *163*, 1660–1672. [CrossRef]
- 84. Prussi, M.; Julea, A.; Lonza, L.; Thiel, C. Biomethane as Alternative Fuel for the EU Road Sector: Analysis of Existing and Planned Infrastructure. *Energy Strategy Rev.* 2021, 33, 100612. [CrossRef]
- 85. EU ETS. EU Emissions Trading System. Available online: https://www.epa.ie/our-services/licensing/climate-change/euemissions-trading-system-/ (accessed on 7 November 2023).
- 86. Mao, C.; Feng, Y.; Wang, X.; Ren, G. Review on Research Achievements of Biogas from Anaerobic Digestion. *Renew. Sustain. Energy Rev.* **2015**, 45, 540–555. [CrossRef]
- 87. Iglesias, R.; Muñoz, R.; Polanco, M.; Díaz, I.; Susmozas, A.; Moreno, A.D.; Guirado, M.; Carreras, N.; Ballesteros, M. Biogas from Anaerobic Digestion as an Energy Vector: Current Upgrading Development. *Energies* **2021**, *14*, 2742. [CrossRef]
- Korres, N.; O'Kiely, P.; Benzie, J.A.H.; West, J.S. Bioenergy Production by Anaerobic Digestion. Using Agricultural Biomass and Organic Wastes; Routledge: London, UK, 2013.
- Atelge, M.R.; Krisa, D.; Kumar, G.; Eskicioglu, C.; Nguyen, D.D.; Chang, S.W.; Atabani, A.E.; Al-Muhtaseb, A.H.; Unalan, S. Biogas Production from Organic Waste: Recent Progress and Perspectives. *Waste Biomass Valorization* 2020, 11, 1019–1040. [CrossRef]
- 90. IEA Outlook for Biogas and Biomethane. Prospects for Organic Growth; World Energy Outlook Special Report; IEA: Paris, France, 2020.
- 91. Ardolino, F.; Cardamone, G.F.; Parrillo, F.; Arena, U. Biogas-to-Biomethane Upgrading: A Comparative Review and Assessment in a Life Cycle Perspective. *Renew. Sustain. Energy Rev.* **2021**, *139*, 110588. [CrossRef]
- 92. Madhusudhanan, A.K.; Na, X.; Boies, A.; Cebon, D. Modelling and Evaluation of a Biomethane Truck for Transport Performance and Cost. *Transp. Res. D Transp. Environ.* **2020**, *87*, 102530. [CrossRef]
- Directive (EU) 2018/850 of the European Parliament and of the Council of 30 May 2018 Amending Directive 1999/31/EC on the Landfill of Waste. 2018. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex:32018L0850 (accessed on 22 January 2024).
- Directive (EU) 2018/851 of the European Parliament and of the Council of 30 May 2018 Amending Directive 2008/98/EC on Waste. 2018, pp. 1–32. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex:32018L0851 (accessed on 21 January 2024).
- 95. Royal Decree 646/2020 of 7 July Regulating the Disposal of Waste by Landfill. 2020. Available online: https://www.boe.es/ buscar/act.php?id=BOE-A-2020-7438 (accessed on 21 January 2024).
- 96. Legislative Decree 1/2009, of 21 July 2009, Approving the Consolidated Text of the Law Regulating Waste. Decreto Legislativo 1/2009, de 21 de Julio, por el que se Aprueba el Texto Refundido de la Ley Reguladora de los Residuos. Available online: https://www.boe.es/eli/es-ct/dlg/2009/07/21/1/con (accessed on 21 January 2024).
- 97. Law 5/2020 of 29 April on Fiscal, Financial, Administrative and Public Sector Measures and Creation of a Tax on Installations That Have an Impact on the Environment. 2020. Available online: https://www.boe.es/buscar/doc.php?id=BOE-A-2020-5569 (accessed on 22 January 2024).
- 98. PREMET25 Metropolitan Programme for the Prevention and Management of Resources and Municipal Waste 2019–2025; Programa Metropolità de Prevenció i Gestió de Recursos i Residus Municipals 2019–2025. 2019. Available online: https: //www.premet25.cat/ (accessed on 22 January 2024).

- 99. Royal Decree 376/2022 of 17 May, which Regulates the Criteria for Sustainability and Reduction of Greenhouse Gas Emissions from Biofuels, Bioliquids and Biomass Fuels, as Well as the System of Guarantees of Origin of Renewable Gases. 2022. Available online: https://www.boe.es/buscar/doc.php?id=BOE-A-2022-8121 (accessed on 22 January 2024).
- 100. PEMAR National Waste Management Framework Plan 2016–2022; PEMAR. Plan Estatal Marco de Gestión de Residuos 2016–2022. Ministerio de Agricultura, Alimentación y Medio Ambiente. 2015. Available online: https://www.miteco.gob.es/content/dam/ miteco/es/calidad-y-evaluacion-ambiental/planes-y-estrategias/pemaraprobado6noviembrecondae_tcm30-170428.pdf (accessed on 22 January 2024).
- 101. Royal Decree 413/2014, of 6 June, Which Regulates the Electricity Production from Renewable Energy Sources, Cogeneration and Waste. 2014. Available online: https://www.boe.es/eli/es/rd/2014/06/06/413/con (accessed on 22 January 2024).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.